Author’s response to reviews

Title: Solitary Fibrous Tumor of the Scrotum: A Case Report and Review of the Literature

Authors:

Tsung-Hsin Chang (b101097063@tmu.edu.tw)

Marcelo Chen (marcelo@mmh.org.tw)

Chih-Chiao Lee (KeithChang0910@gmail.com)

Version: 4 Date: 16 Dec 2019

Author’s response to reviews:

Hu Han (Reviewer 1):
1. In table 1, I can't find the corresponding references with Line 7 (Hu SB et al). There are the same references both Line 7 and Line 8 in Table 1. Please check it. Please check your references carefully. Please follow the journal guidelines and revise reference as necessary!
A: The corresponding references have been corrected. Thanks you for your noticing.

Tarik Benidir (Reviewer 2):
1. Needs to correct typo's and a few spelling mistakes otherwise he has answered my concerns and comments.
A: Thank you for your comment. Spelling mistakes have been checked again.